Author’s response to reviews

Title: ASSOCIATION BETWEEN CARDIORESPIRATORY FITNESS AND METABOLIC RISK FACTORS IN A POPULATION WITH MILD TO SEVERE OBESITY

Authors:
Kathy Do (kathyydo@gmail.com)
Ruth Brown (ruthieemilybrown@gmail.com)
Sean Wharton (wharton.sean@gmail.com)
Chris Ardern (cardern@yorku.ca)
Jennifer Kuk (jennkuk@yorku.ca)

Version: 2 Date: 11 Jan 2018

Author’s response to reviews:

Dear Editor,

Thank you for accepting our manuscript. Please see the revised clean manuscript attached. Please see my point by point responses to your comments below.

Regards,

Jen Kuk
1. Please ensure that all authors are listed separately in the author contributions section. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors

Response - We have now revised the author contributions section to read as follows:

“Authors' Contributions: KD, REB, SW, CIA and JLK made substantial contributions to conception and design. JK and MR were responsible for the analysis and interpretation of data. KD, REB, SW, CIA and JLK were involved in revising the manuscript critically for important intellectual content; and gave final approval of the version to be published. Each author takes responsibility for the content; and agrees to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.” (page 11-12)

2. Please note that an 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. According to the ICMJE guidelines, to qualify as an author one should have:

a) Made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; AND

b) Been involved in drafting the manuscript or revising it critically for important intellectual content; AND

c) Given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; AND

d) Agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Anyone listed as an author must be included in this section. If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship.

Anyone who contributed towards the article who does not meet the criteria for authorship can be acknowledged in the ‘Acknowledgements’ section.
Response – We certify that all authors meet the outlined criteria. There is no change in authorship.

3. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please edit this section to state “Not Applicable” in this section.

Response – Done

4. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response – We have revised the funding section to read as follows.

“Funding: JLK and CIA received funding from CIHR. KD received graduate student funding from the CIHR grant. CIHR had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. SW is the founder and employee of WMC and contributed as an author to the study.”

5. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

Response – Done

6. Please include a key words section under the abstract. This should include three to ten keywords representing the main content of the article.

Response – Done

7. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response – Done